Author’s response to reviews

Title: High-viscosity bone cement for vertebral compression fractures: a prospective study on intravertebral diffusion and leakage of bone cement

Authors:

meiyong wang (wanglei10008@sina.com)
qunhua jin (11083861@qq.com)

Version: 1 Date: 17 Jun 2020

Author’s response to reviews:

1. Introduction
Delete some content.

2. Methods
Underlined parts are supplementary contents.

3. Results
The title has been modified.There is no point in comparing between A&B, also B&C.

4. Discussion
Reference has been modified

5. Limitations
Underlined parts are supplementary contents.

6. Conclusions Underlined parts are supplementary contents.

7. References
Reference has been modified

8. Abstract
The underlined part is the modified content

9. Level of Evidence or Clinical Relevance is not mentioned

10- Keywords
The underlined part is the modified content

11. Title
IThe underlined part is the modified content
12. Tables

The underlined part is the modified content.

13. The underlined part is the modified content